Author's response to reviews

Title: The efficacy of intravenous paracetamol versus dipyrone for postoperative analgesia after day-case lower abdominal surgery in children with spinal anesthesia: a prospective randomized double-blind study

Authors:

Esra Caliskan (esra_ertr@yahoo.com)
Mesut Sener (mesutsenertr@yahoo.co.uk)
Aysu Kocum (akocum@gmail.com)
Nesrin Bozdogan (ecaliskan@baskent-adn.edu.tr)
Semire Serin Ezer (semireserin@yahoo.com)
Anis Aribogan (aaribogan@yahoo.com)

Version: 4 Date: 17 August 2013

Author's response to reviews:

Dear
Ms Marielette Costoy

Thank you for your valuable contributions to our manuscript titled “The efficacy of intravenous paracetamol versus dipyrone for postoperative analgesia after day-case lower abdominal surgery in children with spinal anesthesia: a prospective randomized double-blind study” (Ms. Ref. No: 2676644709851535). The revisions that have been made according to the critiques of the consultants are attached explaining where the changes have been made in the text.

Thank you for your time and consideration.

Best regards,
Esra Caliskan, MD